
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Fri Aug  2 16:11:26 GMT 2013
Year: 13  Doy: 214
Observer: berkey
Thu Aug  1 16:20:03 GMT 2013    CHIP     Start Patrol
Thu Aug  1 16:20:49 GMT 2013    CHIP     End Patrol
Thu Aug  1 16:27:55 GMT 2013    CHIP     Start Patrol
WEATHER COMMENT: Fri Aug  2 16:30:15 GMT 2013
temp 43f, no wind, thin cirrus
___end___
Fri Aug 2 16:32:47 GMT 2013: PSPT Start Patrol
Fri Aug 2 16:47:49 GMT 2013: PSPT Abort Patrol
Fri Aug 2 16:47:55 GMT 2013: PSPT Abort Patrol
Fri Aug 2 16:48:01 GMT 2013: PSPT Abort Patrol
Fri Aug 2 16:48:07 GMT 2013: PSPT Abort Patrol
Fri Aug 2 17:04:40 GMT 2013: PSPT Start Patrol
Fri Aug  2 17:31:54 GMT 2013    MKIV    Start Patrol
Fri Aug 02 17:32:06 GMT 2013 COMP Start Patrol
Fri Aug  2 17:57:04 GMT 2013    MKIV    End Patrol
Fri Aug 02 17:57:16 GMT 2013 COMP End Patrol
Thu Aug  1 18:01:57 GMT 2013    CHIP     LSD
Thu Aug  1 18:03:50 GMT 2013    CHIP     End LSD
Thu Aug  1 18:03:59 GMT 2013    CHIP     BiasLSD
Thu Aug  1 18:04:36 GMT 2013    CHIP     End BiasLSD
Thu Aug  1 18:04:44 GMT 2013    CHIP     Bias
Thu Aug  1 18:05:27 GMT 2013    CHIP     End Bias
Thu Aug  1 18:05:35 GMT 2013    CHIP     ReStart Patrol
Thu Aug  1 20:34:36 GMT 2013    CHIP     End Patrol
NOTE BY BEN: Fri Aug  2 20:43:19 GMT 2013
Dome reconfigured
___end___
Thu Aug  1 20:42:44 GMT 2013    CHIP     Start Patrol
Thu Aug  1 22:18:59 GMT 2013    CHIP     End Patrol
Fri Aug 2 22:19:52 GMT 2013: PSPT Abort Patrol
Fri Aug 2 22:20:00 GMT 2013: PSPT Abort Patrol
NOTE BY BEN: Fri Aug  2 22:24:53 GMT 2013
More clouds in the area closing domes
___end___
NOTE BY BEN: Fri Aug  2 22:45:45 GMT 2013
I will be pulling the dec guider motor assembly out of the spar for modification at HAO next.  

The 22:15:12UT Chip image will be the last ACOS data ever taken.
The ACOS observing mission is now over.  
___end___
NOTE BY BEN: Fri Aug  2 23:53:13 GMT 2013
Dec motor assembly is in a shipping box.



___end___
CoMP COMMENT BY BEN: Sat Aug  3 00:14:43 GMT 2013
CoMP camera off and no nitorgen fill will happen today.
___end___
Sat Aug  3 01:10:13 GMT 2013 
        MkIV    
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